Author’s response to reviews

Title: Correlation analysis between the magnetic resonance imaging characteristics of osteoporotic vertebral compression fractures and the efficacy of percutaneous vertebroplasty: A prospective Cohort Study

Authors:

zhikun li (scoliosis_lizk@126.com)
wei xu (xuweidoc@126.com)
silian wang (WSL@shtrhospital.com)
chao chen (spine_chenchao@126.com)
yifan Li (robie_lee@163.com)
yunhan ji (JYH1715@shtrhospital.com)
xiaodong zhu (scoliosis_zhuxd@126.com)

Version: 2 Date: 23 Jan 2018

Author’s response to reviews:

Thank you for your recognition of my research.